Eco-innovation measurement: A review of firm performance indicators ¹

Eva M. García-Granero^a, Laura Piedra-Muñoz^{a*} and Emilio Galdeano-Gómez^a

^a Department of Economics and Business, University of Almería (Agrifood Campus of International Excellence, ceiA3), Ctra. Sacramento s/n, 04120 Almería, Spain.
Emails: evaggranero@gmail.com; lapiedra@ual.es; galdeano@ual.es
*Corresponding author: lapiedra@ual.es; Tel. +34950015178

Abstract

Increased awareness on sustainability has influenced business organizations to improve their environmental performance and efficiency. In this context, eco-innovation implementation is positioned as a target for organizations to be more sustainable in order to reduce negative externalities and reach governments' green requirements and consumers' demands. The aim of this paper is to provide a critical review of literature on eco-innovation performance indicators. This study identifies the 30 firm performance indicators most cited by researchers and classifies them into four different green innovation types, i.e. product, process, organizational and marketing. A substantial gap has been found throughout the literature on this issue as studies do not include a complete combination of the key performance indicators across the four types of eco-innovation. This information is necessary to obtain an accurate measurement of eco-innovation level and it is useful to companies and stakeholders for performance evaluation. Moreover, understanding which performance indicators are more suitable for measuring the level of environmental innovation affords governments the possibility to draft policies that encourage companies to be more sustainable and firms to implement green practices in a more efficient way.

Keywords: Eco-innovation, performance indicator, literature review, business implementation.

1. Introduction

In recent years, a great deal of research has focused its attention on the impact that the improper use of natural resources has on the environment. This trend, along with the heightened awareness about environmental problems, the limitation of natural resources and the increasing world population, highlights the need to discover new ways of using these resources more efficiently in order to achieve a balance between consumption requirements and sustainability.

¹ Version accepted for publication. 0959-6526/© 2018 Elsevier Ltd. All rights reserved. Published in Journal of Cleaner Production, 191 (2018), 304-317. <u>https://doi.org/10.1016/j.jclepro.2018.04.215</u>. Link: https://www.sciencedirect.com/science/article/abs/pii/S0959652618312514

According to the OECD (2012), the world population will surpass 9,000 million in 2050. Thus, at a time when it will be necessary to increase production of food and other products, topical problems like global warming, deforestation, water pollution, biodiversity loss, excessive generation of waste, and the use of chemical substances will imply a decrease in both productivity and the availability of goods and services. In this context, firms and industries receive special attention as they are considered to contribute most to perpetuating these problems, yet they have the capacity to provide appropriate solutions instead (Remacha, 2017). However, in order to do so, new environmental-friendly production methods as well as improvements in product characteristics, organizational capabilities and marketing practices are required to achieve greater respect for the environment. This objective can be reached by encouraging firms and countries to implement eco-innovations, especially in sectors with considerable environmental impacts in terms of pollution and water and energy consumption, such as agriculture (FAO, 2017).

These innovations, also known as green innovations or environmental innovations, are attracting increasing interest among researchers as a key factor for achieving economic, social and environmental objectives (Läpple et al., 2015). Defining eco-innovation (EI) is not an easy task, although several authors address this topic. According to Kemp and Pearson (2007, p.7), EI is "the production, assimilation or exploitation of a product, production process, service or management or business method that is novel to the organization (developing or adopting it) and which results, throughout its life cycle, in a reduction of environmental risk, pollution and other negative impacts of resources use (including energy use) compared to relevant alternatives". Oltra and Saint Jean (2009, p.1) defined it as "innovations that consist of new or modified processes, practices, systems and products which benefit the environment and so contribute to environmental sustainability", while Kemp and Arundel (1998) and Rennings and Zwick (2003) define environmental innovations as new and modified processes, equipment, products, techniques and management systems that avoid or reduce harmful environmental impacts. For Fussler and James (1996), eco-innovation is the process of developing new products, processes or services which provide customer and business value but significantly decrease environmental impact. Other definitions are also found in works such as Carrillo-Hermosilla et al. (2010), Jänicke (2012) and Tamayo-Obergozo et al. (2017). But, the discussion relative to the definition of EI not only concern researchers, also world organizations discuss this topic. The Eco-Innovation Observatory (2012, p.8) considers it the "introduction of any new or significantly improved product, process, organizational change or marketing solution that reduces the use of natural resources and decreases the release of harmful substances across the whole life-cycle." In the case of the European Commission (2013, p.4), "eco-innovation projects will therefore aim to produce quality products with less environmental impact, whilst innovation can also include moving towards more environmentalfriendly production processes and services. Ultimately, they will contribute towards the reduction of greenhouse gases or the more efficient use of various resources." The Oslo Manual (OECD, 2005) defines innovation as the implementation of a new or significantly improved product (good or service) or process, a new marketing method, or a new organizational method in business practice. According to Europa INNOVA (2006), eco-innovation is the creation of novel and competitively priced goods, processes, systems, services, and procedures designed to satisfy human needs and provide a better quality of life for all, with a minimal life-cycle use of natural resources (materials including energy, and surface area) per unit output, and a minimal release of toxic substances. Furthermore, it is necessary to mention that each author and organization considers different points of view, but all the definitions include two main effects of EI (Hojnik and Ruzzier, 2016b): fewer adverse effects on the environment and more efficient use of resources. These common issues are taken into consideration in this study as the eco-innovation concept.

A wide range of studies on eco-innovation concepts, consequences and drivers have been published, primarily because EI is commonly believed to play a key role in the quest for greater efficiency and sustainability. Nevertheless, studies on its implementation are rather scant (Kemp, 2009). Implementation refers to realization for use according to a European project entitled "Measuring eco-innovation (MEI)" (Kemp and Pearson, 2007, p.7). Therefore, this study focuses on those indicators that measure the implementation of eco-innovations in economic activity. A great deal more of comprehensive research on EI implementation is considered essential in order to identify those eco-innovation performance indicators (EIPI) which allow it to be efficiently measured. In consequence, it would promote progress towards the constitution of a body of knowledge that facilitates not only companies but also governments to implant environmental plans that ensure higher sustainability. As Triguero et al. (2013) mention, a lack of effectiveness of environmental regulation exists. For this reason, a change in the current regulatory framework is needed to enhance EI because environmental regulations play an important role in stimulating EI and combating negative environmental externalities (Ekins, 2010; Demirel and Kesidou, 2011).

The main aim of this article is to offer an overview of the key performance indicators which measure EI at firm level, particularly from the product, process, organizational and marketing perspectives, according to the classification introduced by Macron et al. (2017). To this end, we review the academic literature on EIPI utilizing 104 full articles. No studies were found which provided a comprehensive analysis of the subject from the four EI perspective types. Thus, the paper contributes to the literature in three ways. Firstly, this study provides an academic contribution. As Cooper and Edgett (2008) and Ehrenfeld (2008) note, you cannot manage what you do not measure. In this line, this study offers an overview of key EIPI, contributing to develop a body of knowledge to analyze the level of EI implementation from the point of view of product, process, organizational and marketing perspectives and helping to fill the existing gap concerning this subject. Furthermore, an overview of EIPI makes it

possible to create compound indicators for measuring level of environmental innovation and, subsequently, comparing said levels between countries, sectors or companies (Angelo et al., 2012). Secondly, providing a set of EIPI is a useful base for managers to know which of them should be used to evaluate its performance and diagnose in which EI perspective improvements could be introduced to reduce negative externalities and at the same time add more environmental value and provide a competitive advantage. Finally, due to the fact that EI policies require a holistic view according to Cheng et al. (2014), the current study helps to understand the possible performance indicators for implementing EI.

The remainder of this paper is organized as follows. Section 2 explains the research method and the lines of research used to find and select the publications analyzed. Section 3 shows a descriptive analysis of the findings highlighting the evolution of the research on this subject. In addition, this section analyzes the countries, journals and sectors in which the topic is most widely discussed. Next, Section 4 presents the discussions and contains reviews on EIPI, grouping them into four types of green innovation. This section also introduces a set of key EIPI. Finally, Section 5 concludes the study, discussing the main findings and giving suggestions for future research.

2. Research method

The manner by which EI is measured is evaluated to identify the potential performance indicators necessary for achieving greener and more sustainable procedures. Thus, a systematic literature review has been carried out following the methodology suggested by Tranfield et al. (2003). This approach is also in line with the previous systematic reviews on eco-innovation (e.g., De Medeiros et al., 2014, De Jesús Pacheco et al., 2017; or Hojnik and Ruzzier, 2016b).

This method was chosen as it makes it possible to include large amounts of information contributing to provide a comprehensive view of the field for researchers, answers questions regarding this specific topic and discover new opportunities for future research (De Jesús Pacheco et al., 2017). Furthermore, a systematic review effectively provides a practical perspective as an overall view of EIPI, contributing to create a body of knowledge on this subject in order to determine how to implement ecological practices and policies in the future.

The methodology followed for the literature review included two main phases: Firstly, the extraction and selection of publications in the desired areas; and, secondly, the analysis of the publications retrieved to identify key EIPI. In particular, the systematic literature review followed a five-step scheme according with Tranfield et al. (2003) that included: (i) problem definition; (ii) selection of sources; (iii) selection of studies; (iv) critical appraisal and evaluation; and (v) synthesis.

First, the problem is defined: in line with the overall objective of the research, the aim of the systematic

review was to identify the most cited performance indicators used for measuring the level of EI implementation. Then, the selection of sources and studies is conducted, followed by the description.

Taking into consideration that a systematic literature review must be focused not only on published articles in journals but also on "gray literature" as well (Petticrew and Roberts, 2012), we based the bibliometric analysis on Scopus and Web of Science (WoS) databases in the first stage (Díaz-García et al., 2015; Morioka and de Carvalho, 2016). These databases are considered the most important source of data for scientific research and include titles from Emerald, Elsevier, Springer, Willey, Taylor & Francis or JStor (Bonisoli et al., 2018). Then, a cross-reference analysis and a search in the databases of the main international organizations were conducted with the two-fold aim of analyzing those references that are of interest to the present subject of study and completing the literature review.

Once the literature sources were established, a search based on determining keywords combinations was carried out to select studies. Keywords were selected taking into consideration the main words related to this field and the words most used by researchers. According to Angelo et al. (2012), "environmental innovation" is the term most commonly used in review papers (65%), followed by "eco-innovation term" (22%) and "green innovation" (13%). Therefore, the keywords used for this stage are mainly combinations of the aforementioned terms, along with some eco-innovation implementations. Table 1 presents the keyword combinations used for the search mechanism and the corresponding results for each database. Keyword combinations are reported in rows while databases are reported in columns. The research timeframe covered the period from January 1990 to December 2017.

Key Concept	Search String	Scopus	WoS
Eco-innovation	"Eco-innovation"	382	418
Eco-innovation	"Environmental innovation"	317	314
Eco-innovation	"Green innovation"	173	276
Eco-innovation	"Ecological innovation"	47	58
Eco-innovation	"Measuring innovation" AND "Environment"	12	8
Eco-innovation and product innovation	("Eco-innovation" OR "Environmental innovation" OR "Green innovation") AND ("Product innovation" OR "Product Design" OR ("Product innovation" AND "Recycling materials")	173	145
Eco-innovation and process innovation	("Eco-innovation" OR "Environmental innovation" OR "Green innovation") AND ("Process	66	46

Table 1. Keyword combinations used for the search mechanism and the results for each database.

	innovation" OR "Process efficiency" OR ("Renewable energy" AND "Process improvement")		
Eco-innovation and organizational innovation	("Eco-innovation" OR "Environmental innovation" OR "Green innovation") AND ("Organizational innovation" OR "Organizational change")	32	16
Eco-innovation and marketing innovation	("Eco-innovation" OR "Environmental innovation" OR "Green innovation") AND ("Marketing innovation" OR "Marketing practices")	5	3

The key terms "environmental innovation", "eco-innovation" and "green innovation" embrace an extensive range of sub-topics in spite of being used in conjunction with the four types of EI and also with green practices like "design product", "renewable energy" or "recycling materials". Therefore, search strings were established with the aim of filtering the review and articles being searched. The following fields were selected: Agricultural and Biological Sciences, Environmental Science, Business Science, Economic Science, Ecological Science and Engineering Science. This defined a specific scope for the search and excluded papers whose focus was not relevant to the present study. From this search method, 2.491 papers were found and, as the word combinations were introduced into both databases, 1.969 duplications (79%) had to be removed. Then the title and abstract of each paper were read. Thus, 203 were potentially relevant to this review. After analyzing these complete papers, only those focusing on the EI implementation, i.e. on indicators that measure the implementation of eco-innovations in economic activity, became our set of sources.

The previous procedure led to an initial list of 53 pre-selected articles on eco-innovation implementation. After that, we conducted a cross-reference analysis in order to identify other relevant contributions. Consequently, 51 new references were added.

After analyzing the papers that represent the object of our analysis, the EIPI retrieved were clustered in four different types of EI (product, process, organizational and marketing EI) according with Macron et al. (2017). This classification is described in more detail in the following sections.

3. Results

The selection process described in the previous section yielded a list of 104 publications. This literature review focused on four types of publications. Most of the publications have been classified as journal papers (85), followed by books or book chapters (15) and other related academic publications (4). Table 2 summarizes the range and frequency of the reviewed journals in the field of eco-innovation implementation. A notable 41% (35 articles) of the articles were published in the Journal of Cleaner

Production, and approximately 7% (6 articles) were published in Research Policy. An additional 2% came from the Academy of Management, Journal of Sustainable Development, Journal of Business Ethics, Technovation, Business Strategy and the Environment, Ecological Economics, Packaging Technology and Science, International Journal of Production Economics, Research Technology Management, and Journal of Business Logistics; each respectively contributing 2 articles. Finally, 25 other articles were taken from 25 different journals.

Journal name	Number of articles	Percentage
Journal of Cleaner Production	35	41%
Research Policy	6	7%
Academy of Management Journal	2	2%
Sustainable Development	2	2%
Journal of Business Ethics	2	2%
Technovation	2	2%
Business Strategy and the Environment	2	2%
Ecological Economics	2	2%
Packaging Technology and Science	2	2%
International Journal of Production Economics	2	2%
Research Technology Management	2	2%
Journal of Business Logistics	2	2%
Administrative Science Quarterly	1	1%
Environmental Innovation and Societal Transitions	1	1%
Management Service Quality	1	1%
Resources, Conservation and Recycling	1	1%
Policy Sciences	1	1%
Harvard Business Review	1	1%
International Journal of Operations and Production Management	1	1%
Journal of Marketing Channels	1	1%
Journal of Remanufacturing	1	1%
Interfaces	1	1%
California Management Review	1	1%
Futures	1	1%
Strategic Management Journal	1	1%
Dyna	1	1%
Clean Technologies and Environmental Policy	1	1%
Energy Economics	1	1%
Journal of Environmental Economics and Management	1	1%
SAM Advanced Management Journal	1	1%
Energies	1	1%
Environmental and Resource Economics	1	1%
Journal of Economic Literature	1	1%
The Leadership Quarterly	1	1%
Academy of Management Review	1	1%
Sustainability	1	1%

Table 2. Number of articles published in different journals (1990-2017)

Management Decision	1	1%
Total	85	

An analysis was then conducted in order to determine the main areas of research, the years with the most studies published, and the countries on which most literature is focused (Dangelico, 2015; Caldera et al., 2017).

Over the past two decades, EI has been addressed from different perspectives with the main aim of understanding the motivation for its implementation and how it could be promoted. It should first be noted that studies on this subject have focused on the main factors that prompt firms to innovate in this field. These factors are called "drivers". Research on this topic presents and describes the various dimensions that characterize EI. In contrast, the most recent articles focus on the indicators which measure EI in different sectors and countries (Cheng and Shiu, 2012). Figure 1 shows how the number of publications on the environmental innovation field has significantly increased, up to four times since 2007. This result emphasizes the relatively novel interest on this field of research and the increasing attention that it is receiving. Specifically, in the year 2015 there is a high point in the number of publications due to an increase in studies about which factors motivate the introduction of green practices and about the analysis of eco-innovation impact at environmental and firm levels, particularly in the Journal of Cleaner Production and Innovation Management Policy and Practices. The figures 2 and 3 below display the countries and sectors that have caused this increase in publications.



Figure 1. Eco-innovation publications by year (1990-2017)

Figure 2 shows the geographic distribution and number of articles by countries, analyzing the literature published in this field since 2007, the year in which the number of publications about EI began to increase considerably. The graph shows that Spain is the country with the highest number of publications, followed by the United Kingdom, Italy, France and China.



Figure 2. Eco-innovation publications by country (2007-2017)

Moreover, Figure 3 displays the main subject areas of EI studies. The field with the most research is Business and Management (42.1%), followed by Engineering (29.9%) and Social Science (22.1%). This distribution of publications could indicate that the research findings were also likely applied to the industrial and energy sectors. In contrast, the green innovation field receives scant attention in Agricultural literature (4.4%), particularly when we consider how closely linked this sector is to the environment.



Figure 3. Eco-innovation publications by subject area (2007-2017)

Analyzing Figures 2 and 3, we can come to the follow conclusion. The countries with a greater number of publications on EI are those with a business network constituted by small and medium size firms. In this context, it is evident that they focus their EI studies on the Business and Management sector, making this sector the main subject area for EI papers. However, countries like Spain, Italy or France have a strong agricultural economic sector which is the engine of the economy in many of its regions. Thus, more studies about this sector would be necessary taking into account the considerable impact that the agriculture has on the environment and its close relationship with the use of natural resources.

4. Discussion: Overview of research on eco-innovation performance indicators

Key findings from the systematic literature review are detailed below. The findings emphasize 30 key EIPI. In this study they were clustered into four groups, as was shown in Table 3: (i) product innovation; (ii) process innovation; (iii) organization innovation; and (iv) marketing innovation. Thus, this Section is structured into four parts, one for each type of EI, as it has been mentioned in previous sections. To construct this classification, we followed the review by Macron et al. (2017, p. 84). According to this work, product innovations (i) "can take the form of major or minor changes in the material used, in the technical specification and in the characteristics of the product or service"; process innovations (ii) "are intended to reduce costs, increase quality and provision of the products or services and include improved

techniques in auxiliary support activities"; organizational innovations (iii) "refer to new or significantly improved routines, business models, methods and actions that change firms' practices, relations and decisions"; and marketing innovations (iv) "can occur through changes in product design, product placement, communication, new methods of product delivery, promotion or pricing strategies. Moreover, significant changes in product packaging are also considered important marketing innovations".

Table 3 presents a set of key EIPI retrieved from the analysis of the papers selected in Section 2. The articles were included in at least one category, and some articles are included in more than one. For example, Rodríguez and Wiengarten (2017) was considered to correspond to three different types of eco-innovation (EI), i.e. product, process and organizational, and, accordingly, this reference appears linked to these three types in the classification.

Eco-innovation	Eco-innovation	References	
Types	performance indicators		
		Theyel (2000)	
	Use new cleaner material or	Eder (2003)	
	new input with lower	BID (2007)	
	anvironmental impact	Crabbé et al. (2013)	
	(1 1)	Doran and Ryan (2016)	
	(1.1)	Sierra-Pérez et al. (2016)	
		Castellacci and Lie (2017)	
		Rodríguez and Wiengarten (2017)	
		Van Hemel and Cramer (2002)	
	Use of recycled materials	Cheng and Shiu (2012)	
	(1.2)	Dalhamar (2015)	
		Marcon et al. (2017)	
	Reduce/optimize use of raw	Eder (2003)	
	materials	Hellström (2007)	
	(1.3)	Pigosso et al. (2010)	
		Crabbé et al. (2013)	
Product		Hellström (2007)	
Eco-innovation	Reduce number of product	Cheng and Shiu (2012)	
(1)	components	Doran and Ryan (2016)	
	(1.4)	Castellacci and Lie (2017)	
		Rodríguez and Wiengarten (2017)	
	Eliminate dirty components	Eder (2003)	
	(1.5)		
		Van Hemel and Cramer (2002)	
		Hellström (2007)	
	Product with a longer life	Asif et al. (2012)	
	cycle	Ye and Zhang (2013)	
	(1.6)	Bakker et al. (2014)	
		Dalhamar (2015)	
		Aziz et al. (2016)	
		Garrod and Chadwick (1996)	
	Product ability to be	Bakker et al. (2014)	
	recycled	Dalhamar (2015)	
	(1.7)	Castellacci and Lie (2017)	
		Rodríguez and Wiengarten (2017)	
Process	Reduce chemical waste	Theyel (2000)	

Table 3. Eco-innovation key performance indicators analyzed by the literature.

Eco-innovation	(2.1)	
(2)	Reduce use of water (2.2)	Alkaya and Demirer (2015) Azad and Ancev (2014) Piedra-Muñoz et al. (2018)
	Reduce use of energy (2.3)	Van Hemel and Cramer (2002) Cheng and Shiu (2012) Alkaya and Demirer (2015) Doran and Ryan (2016) Castellacci and Lie (2017) Rodríguez and Wiengarten (2017)
	Keep waste to a minimum (2.4)	Shrivastava (1996) Norberg-Bohm (1999) Cheng and Shiu (2012)
	Reuse of components (2.5)	Hellström (2007) Dalhammar (2015)
	Recycle waste, water or materials (2.6)	Van Hemel and Cramer (2002) Cheng and Shiu (2012) Doran and Ryan (2016) Castellacci and Lie (2017) Podefguez and Wiegerton (2017)
	Environmental-friendly technologies	Garrod and Chadwick (1996) Frondel et al. (2008) Guziana (2011)
	Renewable energy (2.8)	Johnstone et al (2010) Lacerda and Van den Bargh (2014) Nesta et al. (2014) Nicolli and Vona (2016)
	R&D (2.9)	Cohen and Levinthal (1990) Florida (1996) BID (2007) Kemp and Pearson (2008) Cainelli et al. (2015) Rodríguez and Wiengarten (2017)
	Acquisition of machinery and software (2.10)	BID (2007) Kesidou and Demirel (2012) Cainelli et al. (2015) Rodríguez and Wiengarten (2017)
	Acquisition of patents and licenses (2.11)	Griliches (1990) Lanjow and Mody (1996) Jolly and Phillpot (2004) Oltra et al. (2008) Johnstone et al. (2010) Kesidou and Demirel (2012) Cainelli et al. (2015) Rodríguez and Wiengarten (2017)
Organizational Eco-innovation (3)	Green human resources (3.1)	Amabile et al. (1996) Anderson (1998) Andriopoulos (2001) Halbesleben et al. (2002) Naffziger et al. (2003) O'Connor and Ayers (2005) BID (2007) Kemp and Pearson (2008) Montalvo (2003, 2008) Boons and Lüdeke-Freund (2013) Cheng and Chang (2013)
	Pollution prevention plans	Tseng et al. (2013) Hojnik and Ruzzier (2016a) Peng and Liu (2016) Rajala et al. (2016) Frosch and Gallopoulos (1992)
	• • •	▲ `` /

	(3.2)	Tibbs (1992)
		Kemp and Pearson (2008)
	Environmental objectives	Williams et al. (1993)
	(3.3)	
		Baram and Partan (1990)
		Garrod and Chadwick (1996)
		Hamner (2006)
		BID (2007)
	Environmental audit	Kemp and Pearson (2008)
	(3.4)	Montalvo (2003, 2008)
		Eltaveb (2009)
		$\mathbf{Z}_{\text{ailani et al.}}(2007)$
		Boons and Lüdeke Freund (2013)
		Dol Prío and Junguero (2002)
	Environmental advisory	Def Brio and Junquera (2005) PID (2007)
	Environmental advisory	BID(2007)
	(3.5)	Scarpellini et al. (2012)
		De Jesús Pacheco et al. (2016)
	Invest in research	Porter and Van der Linder (1995)
	(3.6)	Horbach (2008)
		Cramer et al. (1991)
		Frosch and Gallopoulos (1992)
		Cramer and Schot (1993)
		Frosch (1994)
		Florida (1996)
		Anderson (1998)
		Becker and Dietz (2004)
		Hamner (2006)
		Chen (2008)
	Cooperation with stakeholders	Eltaveh (2000)
		De Marchi (2007)
	(3.7)	Motos and Silvestre (2012)
		Matos and Silvestie (2013)
		Segarra-Ona and Peiro-Signes
		(2014)
		Ghisetti and Reinnings (2014)
		Ghisetti et al. (2015)
		Ghisetti and Pontoni (2015)
		Roscoe et al. (2015)
		Bossle et al. (2016)
		Rodríguez and Wiengarten (2017)
	Namenalasta	Blättel-Mink (1998)
	inew markets	Niinimäki and Hassi (2011)
	(3.8)	Loorbach and Wijsman (2013)
		Stock (1992)
		Blättel-Mink (1998)
		Carter and Ellram (1998)
	New systems	Moore (2005)
	(remanufacturing systems	Fl Korchi and Millet (2011)
	and transport systems)	$\Delta \operatorname{sif}$ et al. (2012)
	(3.9)	$\mathbf{X}_{\mathbf{A}} \text{ and } \mathbf{Z}_{\mathbf{A}} \mathbf{G}_{\mathbf{A}} \left(\mathbf{Z}_{\mathbf{A}} \mathbf{Z}_{\mathbf{A}} \right)$
		$1 \in \text{and } \text{Lang} (2013)$
		Dakker et al. (2014)
		<u>Intani et al. (2014)</u>
		Stock (1992)
		Hart (1995)
	Returnable/Reusable	Shrivastava (1995)
		Rosenau et al. (1996)
Marketing		Carter and Ellram (1998)
Eco-innovation	Packaging	Rogers and Tibben-Lemke (1998)
(4)	(4.1)	Christmann (2000)
		Duhaime et al. (2001)
		Van Hemel and Cramer (2002)
		Twede and Clarke (2005)
		Zalani et al. (2012)
		\square

_

		Silva et al. (2013)
Green Design Packaging (4.2)	Green Design Packaging	Löfgren (2005) Martin et al. (2006) Henriksson et al. (2009) Langley et al. (2011) Cheng and Shiu (2012) Juul (2012)
		Zailani et al. (2012) Plumb et al. (2013) Wever and Vogtländer (2014) Lindh et al. (2016) Wilkström et al. (2016)
	Quality certifications (4.3)	Hamner (2006) Eltayeb (2009) Chiarvesio et al. (2015) Li and Hamblin (2016)

The set of key EIPI established in Table 3 highlights the performance indicators most cited by the EI literature to analyze and measure the EI in different sectors and countries, offering a state of art in this topic. It is discussed in the following section.

4.1. Product eco-innovation

The materials used to make a product as well as the product characteristics themselves have an impact on the environment. Thus, numerous research studies on how to implement environmental innovations have focused on improving the type and quality of inputs and product sustainability in order to reach current environmental requirements and to decrease negative externalities. More specifically, this study of the literature found 7 EIPI based on EI products (1).

Related to the determinants of the product's characteristics and, in turn, its environmental impact, the literature enhances the inputs used to make a product as one of the major points to have in consideration to implant EI. In this sense, reducing the use of dirty inputs (1.5) or substituting them for cleaner or less polluting materials (1.1) contributes to decreasing waste and CO² emissions. The materials used to make a product comprise one of the EIPI that a great deal of research highlights as one of the factors necessary for creating products that are more environmental-friendly.

Some authors emphasize the importance of reducing or optimizing the use of raw materials (1.3) to obtain products (e.g., Eder, 2003; Hellström, 2007; Crabbé et al., 2013). The utilization of raw materials as an input in product manufacturing has a significant negative impact on the environment for two reasons. Firstly, its consumption ultimately increases (Agrawal and Ülkü, 2011). Secondly, the decarbonation of raw materials increases carbon dioxide (CO²) emissions (Ishak et al., 2016). Thus,

reducing the use of raw materials (1.3) by a sector or a company is a performance indicator that should be taken into consideration for measuring EI and sustainability level. In this line, Pigosso et al. (2010) support products whose raw materials are obtained from other products as a way to reduce contaminants. Also, Eder (2003) focuses attention on the necessity of substituting raw materials (1.3) for cleaner alternatives.

Using new cleaner materials or new inputs with lower environmental impact (1.1) is also used in EI literature as an indicator of a product's level of efficiency (e.g., Dora and Ryan, 2016; Castellacci and Lie, 2017; Rodríguez and Wiengarten, 2017). Theyel (2000) highlights the necessity of using cleaner or less polluting materials in a review based on the plastics and resins sector and the ink manufacturing sector in the US chemical industry; while Crabbé et al. (2013), in a study on Flemish production firms, emphasize the importance of innovating to obtain sustainable materials which contribute to making products more respectful of the environment. Also, BID (2007) recognizes the use of new sustainable materials (1.1) as an indicator of the innovation effort of a company. Other empirical market studies note the importance of using cleaner materials (1.1) to reduce the negative environmental impact of firms. According to Sierra-Pérez et al. (2016), introducing the use of cork to replace non-renewable materials in the construction sector decreases ecological impact. Eder's research (2003) highlights eliminating the use of dirty or polluting components (1.5) to make a product in order to obtain fewer contaminant products.

Other performance indicators related to the inputs used in manufacturing show the improvement in product efficiency. One of these indicators is the use of recycled inputs (1.2). According to Dalhamar (2015) and Marcon et al. (2017), the use of recycled materials (1.2) is an essential performance indicator of green innovation. In accordance with this concept, Van Hemel and Cramer (2002) and Cheng and Shiu (2012) emphasize that the use of recycled product components (1.2) is another tool for manufacturing more sustainably. Similarly, research conducted by Hellström (2007) highlights the reduction of the number of product components (1.4) as another successful indicator of product EI. Along this line, other authors introduce the indicator "reduce material per unit of output" (1.4) in the research to measure the level of efficiency of a product (e.g., Doran and Ryan 2016; Castellacci and Lie, 2017; Rodríguez and Wiengarten, 2017).

Analyzing the EI literature is demonstrated the product characteristics are strongly correlated with environmental impact. A product's durability (1.6) and ability to be reused (1.7) are the two most relevant characteristics studied by the EI literature as they are directly linked to product efficiency, reduced consumption of resources, and lower gas emissions. Hellström (2007), Bakker et al. (2014) and Aziz et al. (2016) present the long-life product (1.6) as an effective tool for obtaining a greater level of environment sustainability. Dalhammar (2015) discusses product durability (1.6) and the technical

guarantees on life cycle as improvements which can provide more environmental efficiency, while Van Hemel and Cramer (2002), in their study which analyzes the environmental performance of the US chemical industry, introduce an investigation to extend product lifetime (1.6) by providing a list of the main solutions for achieving sustainability. Moreover, Asif et al. (2012), Ye and Zang (2013) and Bakker et al. (2014) highlight remanufacturing as a strategy to extend product lifetime (1.6).

Additionally, Dalhammar (2015), Castellacci and Lie (2017) and Rodríguez and Wiengarten (2017) introduce the ability of a product to be recycled (1.7) after use as a key performance indicator for measuring EI level. This practice leads to the reduction of waste as it extends product life at the same time. Along this line, Garrod and Chadwick (1996) carried out a survey of companies located in the South of England to determine how firms had handled the increase of environmental pressures. Their analysis identified several firm performance indicators that were implemented, one which was the recycling of part of the used final product (1.7). Also, Bakker et al. (2014), in their study on household products, emphasize the recycling of products (1.7) as an essential tool in order to achieve greener practices.

4.2. Process eco-innovation

The environmental impact of a company is not only due to what the company produces but also how the company manufactures its products. Groenewegen et al. (1996) have established the relationship between the manufacturing processes of a company and negative environmental impact. Thus, it is necessary to take into consideration improvements in manufacturing processes and include relevant EI indicators in order to efficiently measure levels of environmental innovation. This study of the literature has identified 11 EIPI based on improvements in manufacturing processes (2).

The total use of water or energy is a widely-used method in EI literature for analyzing process improvement. Alkaya and Demirer (2015), in a review of the Turkish chemical industry, use the indicators "reduce water consumption" (2.2) and "reduce energy consumption" (2.3) to study the sustainability of the sector's production processes and whether companies attempt to fulfill green requirements. In the same way, irrigated agriculture is one of the sectors where the use of water has been most analyzed. Many studies measure the effects of eco-innovations aimed at optimizing water usage on farmers' environmental impact (Azad and Ancev, 2014; Piedra-Muñoz et al., 2017, 2018). Other works which follow this line include: Van Hemel and Cramer (2002), who analyze a group of 77 small and medium sized companies (SMEs); Cheng and Shiu (2012), who measure EI from the perspective of implementation; Doran and Ryan (2016), who base their review on the Irish Community Innovation Survey; Catellacci and Lie (2017), who focus their study on manufacturing firms in Korea; and

Rodríguez and Weingarten (2017), who study several German industries and highlight energy reduction (2.3) in the manufacturing process as a performance indicator to measure environmental efficiency.

The level of waste (2.4) in a process is also analyzed as a cause of pollution. Thus, Shrivastava (1996), Norberg-Bohm (1999) and Cheng and Shiu (2012) emphasize the importance of introducing new technologies with the aim of reducing waste to a minimum. Theyel (2000) expands on this reasoning by proposing the idea of reducing chemical waste (2.1) in production processes as much as possible. According to the Van Hemel and Cramer (2002) study, which focused on the US chemical industry, firms that innovate in terms of reducing chemical waste (2.1) are leaders in adopting environmental practices.

Materials-saving is another key performance indicator for measuring EI and the efficiency level of a process. This indicator can be viewed from two perspectives. On the one hand, the reuse of components or materials (2.5) attracts attention as a positive way of being greener in the manufacturing process (Hellström, 2007; Dalhammar, 2015). On the other hand, the recycling of waste, water, materials or inputs (2.6) is another means of reducing negative environmental impact. Thus, some authors introduce the indicator "recycled waste, water and materials" (2.6) in their studies to measure environmental innovativeness (e.g., Doran and Ryan, 2016; Castellacci and Lie, 2017; Rodríguez and Wiegarten, 2017). Furthermore, according to Van Hemel and Cramer (2002) the eco-indicator "recycling of materials" (2.6) is the most successful among firms to improve their environmental performance.

The level of investment carried out by a company is a relevant performance indicator of its effort to be greener. In this context, some authors analyze company investment in patents (2.11) as a means of achieving environmental innovations to improve energy consumption and material efficiency (Kesidou and Demirel, 2012; Cainelli et al., 2015; Rodríguez and Wiengarten, 2017). In a study on renewable technology, Johnstone et al. (2010) identify the number of patents (2.11) as a measurement indicator of EI. Additionally, Griliches (1990), Lanjow and Mody (1996), and Jolly and Phillipot (2004) shows that patents are a good indicator for measuring innovation activity level. Furthermore, the European Comission contemplates the "eco-patents" as an indicator of the level of innovative activity in the environmental field and as a way for studying eco-innovations (Oltra et al., 2008).

Nevertheless, not all company research efforts and investments are always patented. Thus, in addition to the number of patents, other practices exist that this indicator does not take into consideration (Oltra et al., 2008; Artz et al., 2010). For this reason, although the number of patents is strongly correlated with research and development (R&D) spending, it is necessary to include the indicator "number of patents" along with others such as acquisition of machinery and software (2.10) or R&D investments (2.9) to achieve a more accurate view of the innovative reality of a firm. R&D activity (2.9) is treated by some authors as a key performance indicator in the EI process. In fact, it has been shown that firms which

implement R&D activities (2.9) are more likely to be environmentally innovative than firms that are not R&D active since the former have a higher absorptive capacity (Cohen and Levinthal, 1990; Cainelli et al., 2015). According to Florida (1996), firms that are R&D active (2.9) improve their productivity and reduce negative environmental impact. Thus, some authors (e.g., BID, 2007; Kemp and Pearson, 2008; Rodríguez and Weingarten, 2017) introduce the indicator R&D (2.9) to measure EI and subdivide it into internal or external R&D.

In addition, other authors (e.g., Kesidou and Demirel, 2012; Cainelli et al., 2015; Rodríguez and Wiengarten, 2017) focus their studies on the acquisition of machinery (2.10) as a key factor for the purpose of more efficient use of energy and materials. BID (2007) also illustrates the importance of incorporating new capital assets, i.e., hardware and software (2.10), in order to implement ecological innovations in a company.

The use of renewable energy (2.8) and environmental-friendly technologies (2.7) are two more relevant EIPI emphasized by the literature in this field as ways of achieving more efficient manufacturing processes, making them crucial for addressing global environmental aims. Frondel et al. (2008) highlight the environmental benefit of introducing *end-of-pipe* technologies (2.7) in manufacturing processes, whereas Guziana (2011) concludes that clean technologies (2.7) are more proactively innovative than the former. Along this line, Garrod and Chadwick (1996), in their survey of environmental strategies carried out by companies located in the South of England, determined that investment in clean technology (2.7) is a tool that can be implemented to fulfill ecological requirements. Moreover, other articles address the importance of introducing renewable energies (2.8) in company processes in order to improve quality of life for current and future generations and to meet public environmental objectives (e.g., Lacerda and Van den Bergh, 2014; Nesta et al., 2014; Nicolli and Vona, 2016).

4.3. Organizational eco-innovation

Chen (2008) illustrates the importance of the relationship between green intellectual capital and the competitive advantage of firms. The Chen study which focused on the Taiwanese information and electronics industry, emphasizes the positive correlation between these two indicators. According to its findings, there are three types of green intellectual advantage: green human capital, green structural capital, and green relational capital. Furthermore, the study identified 9 EIPI related to organizational eco-innovation (3). Said indicators are introduced below.

Green human capital (3.1) refers to the collective knowledge, skills, creativity, experience and capabilities of employees. In this sense, based on a study of the In-Bond industry in the northern region

of Mexico, Montalvo (2003) highlights the influence of managerial characteristics (3.1) on EI and the environmental-economic risks of developing cleaner technologies and manufacturing processes. Other studies (e.g., Montalvo, 2008; Boons and Lüdeke-Freund, 2013; Chen and Chang, 2013) support this idea arguing that senior staff (3.1) can encourage employees to be more creative, innovative and respectful with the environment. According to Andriopoulos (2001) and Halbesleben et al. (2003), leaders (3.1) with appropriate green perspectives play a key role in facilitating organizational creativity as well as the implementation of environmental innovations. Amabile et al. (1996) highlight creativity (3.1) as a starting point for innovation. Furthermore, Rajala et al. (2016), in a study of the US-based carpet manufacturer Interface, illustrate the role of the managerial agency (3.1) in driving environmentally sustainable practices in a company in order to unite firm culture and firm orientation with a green business model. The relationship between employing managers who are more in tune with environmentally conscious practices and greener business models based on better ecological performance and higher investments in environmental initiatives has also been highlighted by other researchers, e.g., Anderson (1998), O'Connor and Ayers (2005), and Hojnik and Ruzzier (2016a). In this line, Naffziger et al. (2003) and Tseng et al. (2013) establish the relationship between the presence in a company of a manager with a higher level of environmental (3.1) concern and the time and money invested in environmental initiatives. Moreover, Peng and Liu (2016), in a study which explores the determinants of EI, include the indicator "managerial environmental awareness" (3.1) in order to measure green innovation. In addition, BID (2007) and Kemp and Pearson (2008) accentuates the importance of green human resources (3.1) as an indicator which shows the innovative effort of a firm.

Green structural capital includes organizational capabilities, organizational commitments, organizational culture and philosophies, patents, copyrights, etc. Some of these have been analyzed in the previous section as processes of EI. Nevertheless, organizational cultures and philosophies can also be considered an organizational EI. According to Battisti (2008), it is not only important for firms to generate innovations; innovations must be adopted and used by firms, incorporated into their routines and their company philosophy. Thus, environmentally-oriented culture is another green performance indicator that should be taken into account by the literature for measuring EI. In a review carried out by Williams et al. (1993), this indicator, i.e. environmentally-oriented culture, is measured using the number of environmental objectives (3.3) included in production plans and operations. The reviews of Frosh and Gallopoulos (1992), Tibbs (1992) and Kemp and Pearson (2008) also highlight the inclusion of environmental plans (3.2) in production processes.

From the point of view of several researchers (e.g., Baram and Partan, 1990; Hamner, 2006; Zailani et al. 2012), conducting external environmental audits (3.4) is another good performance indicator for measuring the level of company commitment to environmental requirements. In their study based on a

firm survey, Garrod and Chadwick (1996) also introduce environmental audits (3.4) as a growing indicator used to achieve EI. Ecological audits (3.4) provide firms with knowledge as to whether their green innovation is being effective and, depending on the result, firms can implement new ecological practices to reduce their environmental impact. Thus, Kemp and Pearson (2008) enhances auditing systems as a key organizational innovation for the environment. In addition, consulting services (3.5), which ensure compliance with environmental standards, constitute another tool that has the potential to increase the EI level of a company (e.g., del Brio and Junquera, 2003; Scarpellini et al., 2012; de Jesus Pacheco et al., 2016). According to this, BID (2007) enhances the outsource consulting and technical assistance (3.5) as green innovative strategies.

Investment in research (3.6) is another key point that firms should introduce in their corporate culture. Although controlling pollution can be effective, it is not always the most efficient way to satisfy environmental requirements. Therefore, restructuring a firm's approach toward environmental management, from pollution control to pollution prevention, may be the most ecologically-driven method (Gottlieb et al., 1995). Accordingly, investing in research becomes an effective tool for achieving this goal (Porter and Van der Linder, 1995; Horbach, 2008).

Green relational capital is defined as the relationships of the company with customers, suppliers, network members, and partners regarding environmental management and EI. Accordingly, cooperation with stakeholders (3.7) enhances the creation of competitive advantage and simultaneously helps to achieve environmental objectives (e.g., Matos and Silvestre, 2013; Roscoe et al., 2015; Rodríguez and Wiengarten, 2017). According to Cramer et al. (1991), Cramer and Schot (1993) and Frosch (1994), restructuring firm relationships with pressure groups (3.7) is an important factor for obtaining information about the environment and providing assistance to suppliers and customers. Furthermore, forming partnerships with these groups (3.7) affords greater possibilities to seek out solutions to environmental problems (Frosch and Gallopoulos, 1992) and to renew firm business models to make them greener and more sustainable (Anderson, 1998). Florida (1996) and Chen (2008) also highlight the close positive relationship between firm-supplier ties (3.7) and the creation of new environmental improvement opportunities. Cooperation with suppliers, universities and public research institutions (3.7) has three significant benefits. First, it provides the firm with knowledge (e.g., Ghisetti and Reinning, 2014; Ghisetti and Pontoni, 2015; Bossle et al., 2016). Second, it allows a firm to obtain information with the aim of improving products and processes (De Marchi, 2012; Segarra-Oña and Peiró-Signes, 2014). Third, it makes it possible for the firm to develop technological capabilities necessary to generate innovation (Becker and Dietz, 2004; Ghisetti et al., 2015).

One well-known cooperation method (3.7) is to create supplier questionnaires. These surveys provide firms with information about their level of environmental commitment and the quality of their

environmental characteristics, activities and practices. In addition, firms obtain an idea of what kind of image their activities produce in the eyes of stakeholders (Eltayeb, 2009; Hamner, 2006). This practice allows firms to correct non efficient activities and implement new, greener ones.

The development of new market niches (3.8) is considered by some researchers to be another useful tool for the purpose of implementing green innovations (e.g., Blättel-Mink, 1998; Niinimäki and Hassi, 2011; Loorbach and Wiisman, 2013) and introducing new systems (3.9) (Blättel-Mink, 1998). According to El Korchi and Millet (2011), introducing remanufacturing systems or reverse logistic channels (3.9) allows firms to reduce environmental impact by reducing waste and extending product life cycle. Asif et al. (2012) and Ye and Zang (2013) believe multiple life cycle products (MLPs) (3.9) constitute an important strategy for developing sustainable products and that remanufacturing is the best tool for achieving this goal. Additional research supporting remanufacturing systems (3.9) has also been published in a number of other relevant works (e.g. Stock and Lambert, 2001; Moore, 2005; Bakker et al., 2014). Finally, implementation of new transport systems (3.9) based on new routes, short distances, and the replacement of diesel fuel is another means of applying green innovation (Iritani et al., 2014) and achieving less pollution through the reduction of CO² emissions.

4.4. Marketing eco-innovation

Marketing innovation activities are relevant performance indicators for implementing and measuring EI, as stated by BID (2007). However, marketing green innovation has received less attention than other types of EI in environmental literature, which by no means makes it any less important. This review has identified 3 EIPI based on marketing (4). Recently, certain research has focused on identifying the environmental marketing indicators that can measure the level of EI implementation in order to reduce the negative environmental impacts of companies; achieve greater efficiency; and find new ways to carry out ecological innovation in the four dimensions: product, process, organizational and marketing.

Some environmental policies have focused on packaging, for example, the Directive 94/62/EC in the European Union, the response to the large amount of waste disposable packaging generates and its negative environmental impact (González-Torre et al., 2004). Thus, the use of returnable packaging (4.1), which can be recycled and reused, contributes to EI by increasing product efficiency while reducing waste and resource consumption. Some examples of relevant publications on the environmental benefits of using returnable packaging (4.1) include Rogers and Tibben-Lembke (1998), Duhaime et al. (2001) and Twede and Clarke (2005). In this line, Stock (1992), Carter and Ellram (1998) and Silva et al. (2013) focus their studies on the reduction of waste and the improvement in resource efficiency resulting from the use of returnable packaging. Furthermore, Zailani et al. (2012) emphasize

the need for design innovation in reusable packaging in order to enhance sustainability. Similarly, more authors (e.g., Hart, 1995; Shrivastava, 1995; Christmann, 2000) highlight the importance of packaging design (4.2) that can be reused in order to improve the sustainable performance of firms. Other studies agree with this environmental innovation indicator (e.g., Rosenau et al., 1996; Van Hemel and Cramer, 2002). In the literature the importance of packaging design (4.2) to influence consumer interaction with products is demostrated (Löfgren, 2005). Jelsma (2006) illustrates, for example, that product attributes can determine consumer behavior. Some authors (e.g., Zailani et al., 2012; Wever and Vogtländer, 2014; Wilkström et al., 2016) question the importance of including 'sustainable' packaging design as a means to fulfill ecological requirements, discussing whether it encourages customers to reduce food waste and recycle packaging. In order to measure of the extent to which sustainable packaging has been implemented, a great deal of researchers have debated the attributes packaging must possess in order to be green, such as: easy to empty (Langley et al., 2011; Juul, 2012); easy to clean (Langley et al., 2011); easy to separate into different fractions (Henriksson et al., 2009) easy to fold (Martin et al., 2006); provides information about how to sort (Henriksson et al., 2009; Langley, et al., 2011); contributes by extending time between packaging date and expiration date (Plumb et al., 2013; Lindh et al., 2016); and contains the desired quantity (Lindh et al., 2016). According to Cheng and Shiu (2012), simplifying packaging is also a necessary way to obtain sustainable packaging.

Although the main focus in the literature about EI marketing type is on packaging, customer buying decisions are not only influenced by traditional criteria like cost, quality, and delivery but also by green firm image and sustainable firm activities. This is due to the increase in market awareness of environmental problems. In this context, quality certifications (4.3) are the best way to show markets whether a firm is fulfilling environmental requirements. Product certification according to international standards, such as ISO 14001 or Globalgap, is an increasingly necessary requisite for companies wishing to gain entry to numerous markets. This issue has been addressed by various authors, such as Hamner (2006), Eltayeb (2009) and Chiarvesio et al. (2015). Additionally, Li and Hamblin (2016), in a study based on pharmaceutical manufacturing companies in Tianjin (China), introduced the indicator "ISO 14001" to analyze the impact that some factors (CO₂, packaging, waste...) have on cleaner production. In this context, standards certifications related to environmental management can be a good EI performance indicator to measure the efforts to accomplish the environmental requirements.

5. Conclusions and future research

EI implementation has received little attention in comparison with the wide range of studies published on EI concepts, consequences and drivers (Kemp, 2009). Thus, the present study looks to fill the existing

gap, analyzing the literature on key EIPI, and synthesizes the most current research on this topic, adding value in the following ways. On the one hand, it offers an overview of which performance indicators are the most cited in the EI literature. In this line, this review contributes to develop a body of knowledge to analyze and measure the level of EI implementation that can potentially guide recommendations for future economic, social and environmental policies in order to reach current environmental objectives (Carrillo-Hermosilla et al., 2010; Boons and Lüdeke-Freund, 2013). This is particularly interesting because EI policies play a key role in the EI implementation as Rennings (2000), Del Río et al. (2010) and Wagner and Llerena (2013) mention. On the other hand, a set of EIPI was developed to show the most important performance indicators that must be included in the four types of EI (product, process, organizational and marketing), which can also be used as a guide to obtain an efficient environmental innovation measurement. Furthermore, this can be useful to create compound indicators for measuring level of environmental innovation and, subsequently, comparing said levels between countries, sectors or companies.

It is clear that the environmental impact of firms' daily activities such as CO₂ emissions, non-efficient use of resources, and high waste levels of water and energy, increases concern regarding their ecological performance. Thus, the implementation of EI is critically important due to the ever increasing demand for a cleaner environment. In this context, research works in business, environmental and economic literature are focused on trying to measure and analyze EI implementation levels in order to discover how environmental actors can reduce their negative environmental impacts, fulfill green requirements and be more efficient to ensure the well-being of current and future generations. The careful study of literature focused on measuring and analyzing EI implementation in different countries and sectors has generated the following conclusions. It is observed that a large portion of the literature on measuring EI are focused on product, process and organizational EI type. Thus, the 36% of the 30 key performance indicators identified corresponding to process EI, 30% to organizational EI and 23% to product EI. In this sense, marketing EI type has received little attention by the literature in spite of its increasingly known environmental impact. Moreover, the vast majority of literature on EI measurement is focused on the Business, Management and Engineering sector; thus, more studies should be carried out in sectors like agriculture due to its close relationship with the use of natural resources and environmental externalities. Our study also identified some weakness on existing studies on EI measurement. Most research has focused on exploring one or two types of EI (product, process, organizational or marketing EI), but not all four types in specific areas. This fact does not afford an efficient, comprehensive study on EI and, instead, offers a very limited vision of the level of EI in firms, sectors or countries. The most complete studies on this subject have been carried out by Doran and Ryan (2016), Castellacci and Lie (2017) and Rodríguez and Wiengarten (2017). However, they only investigate EI implementation in product and process type, so their conclusions do not accurately reflect the reality of the firm, and they

can only provide a limited idea of the level of ecological innovation implementation. Another notable weakness in existing research on EI implementation is related to the performance indicators that are included. Choosing a complete combination of indicators in each EI type is not an easy task, and evidence suggests that the majority of studies include indicators that are chosen in rudimentary ways, with little attention given to which indicators add more environmental value in each sector and firm. Although some methods are better than others, no single method or indicator is ideal. Different methods should be applied for analyzing eco-innovation, as Kemp (2009, p.103) mentioned: "to see the whole elephant, instead of just a part". Consequently, it would be particularly interesting for future research to conduct studies in which all types of EI, as well as the most relevant green indicators in each type, are included. Future research that applies new questionnaires in different sectors can help to discover new ways of marketing. The inclusion of new indicators would help to fill existing gaps related to those EIPI that have already been identified. This is particularly useful when seeking to obtain an accurate measurement of EI level.

A number of limitations of this study can be cited. Firstly, it follows a strictly theoretical research method based on previous research. Future works could be aimed at studying actual case studies to identify what companies are actually doing. Secondly, another shortcoming is the search frame, as the database choice for the paper search could be expanded. Thirdly, one more limitation is related to the criteria initially used for the paper selection. Expanding criteria could lead to other EIPI not covered by this study. Thus, all these points are also opportunities for future research.

Finally, the results have corroborated that environmental innovation should be analyzed as a whole in order to have a sound method for measuring EI level including the four dimensions of EI (product, process, organizational and marketing) and a complete indicator combination in each type. Looking to the future, this research has provided much information with implications for industry, governments and academia to understand which EI indicators can be implemented by environmental stakeholders to reduce their negative environmental impact and become greener. This study also supplies a set of EI implementation indicators to aid practitioners and policy-makers in assessing the balance between company activities and sustainability. These are relevant opportunities to advance the academic perspective towards the constitution of a body of knowledge on this research topic.

Acknowledgments: This research was partially funded by Spanish MCINN and FEDER aid [projects ECO2014-52268-P and ECO2017-82347-P] and by Andalusian Regional Government [project SEJ-2555, Consejería de Economía, Innovación y Ciencia].

References

- Agrawal, V.V., Ulku, S., 2011. The Role of Modular Upgradability as a Green Design Strategy. Working paper, McDonough School of Business, Georgetown University.
- Alkaya, E., Demirer, G. N., 2015. Reducing water and energy consumption in chemical industry by sustainable production approach: a pilot study for polyethylene terephthalate production. Journal of Cleaner Production 99, 119-128.
- Amabile, T. M., Conti, R., Coon, H., Lazenby, J., Herron, M., 1996. Assessing the work environment for creativity. Academy of Management Journal 39 (5), 1154-1184.
- Anderson, R., 1998. Mid-course Correction. Towards a Sustainable Enterprise: the Interface Model. Chelsea Green Publishing Company, White River Junction, VT.
- Andriopoulos, C., 2001. Determinants of organizational creativity: A literature review. Management Decision 6(3), 240-249.
- Angelo, F. D., Jabbour, C. J. C., Galina, S. V., 2012. Environmental innovation: in search of a meaning.World Journal of Entrepreneurship Management and Sustainable Development 8 (2-3), 113-121.
- Artz, K.W., Norman, P.M., Hatfield, D.E., Cardinal, L.B., 2010. A longitudinal study of the impact of R&D, patents, and product innovation on firm performance. Journal of Product Innovation Management 27 (5), 725-740.
- Asif, F. M. A., Bianchi, C., Rashid, A., Nicolescu, C. M., 2012. Performance analysis of the closed loop supply chain. Journal of Remanufacturing 2, (4), 1-21.
- Azad, M.A.S.; Ancev, T., 2014. Measuring environmental efficiency of agricultural water use: A Luenberger environmental indicator. Journal of Environmental Management, 145, 314–320.
- Aziz, N. A., Wahab, D. A., Ramli, R., Azhari, C. H., 2016. Modelling and optimisation of upgradability in the design of multiple life cycle products: a critical review. Journal of Cleaner Production 112, 282-290.
- Bakker, C., Wang, F., Huisman, J., Hollander, M., 2014. Products that go round: exploring products life extension through design. Journal of Cleaner Production 69, 10-16.
- Baram, M. S., Partan, D. G., 1990. Corporate disclosure of environmental risks: U.S. and European law. Butterworth Legal Publishers (Ed.), Boston.
- Battisti, G., 2008. Innovations and the economics of new technology spreading within and across users: gaps and way forward. Journal of Cleaner Production 16S1, S22-S31.
- Becker, W., Dietz, J., 2004. R&D cooperation and innovation activities of firms-evidence for the German manufacturing industry. Research Policy 33, 209-223.
- BID (Banco Interamericano de Desarrollo), 2007. Módulo de capacitación para la recolección y el análisis de indicadores de innovación. Working Paper 8.

- Blättel-Mink, B., 1998. Innovation towards sustainable economy- the integration of economy and ecology in companies. Sustainable Development 6 (2), 49-58.
- Bonisoli, L., Galdeano-Gómez, E., Piedra-Muñoz, L., 2018. Deconstructing criteria and assessment tools to build agri-sustainability indicators and support farmers' decision-making process, Journal of Cleaner Production, 182, 1080-1094, https://doi.org/10.1016/j.jclepro.2018.02.055.
- Boons, F., Lüdeke-Freund, F., 2013. Business models for sustainable innovation: state-of-the-art and steps towards a research agenda. Journal of Cleaner Production 45, 9-19.
- Bossle, M. B. Dutra De Barcellos, M., Vieira, L. M., Sauvée, L., 2016. The drivers for adoption of ecoinnovation. Journal of Cleaner Production 113, 861-872.
- Cainelli, G., De Marchi, V., Grandinetti, R., 2015. Does the development of environmental innovation require different resources? Evidence from Spanish manufacturing firms. Journal of Cleaner Production 94, 211-220.
- Caldera, H. T. S., Desha, C., Dawes, L., 2017. Exploring the role of lean thinking in sustainable business practice: A systematic literature review. Journal of Cleaner Production, http://dx.doi.org/10.1016/j.clepro.2017.05.126
- Carter, C. R., Ellram, L. M., 1998. Reverse logistics: a review of the literature and framework for future investigation. International Journal of Business Logistics 19, 85-102.
- Carrillo-Hermosilla, J., del Río, P., Könnölä, T., 2010. Diversity of eco-innovations: reflections from selected case studies. Journal of Cleaner Production 18 (10-11), 1073-1083.
- Castellacci, F., Lie, C. M., 2017. A taxonomy of green innovators: empirical evidence from South Korea. Journal of Cleaner Production 143, 1036-1047.
- Chen, Y., 2008. The positive effect of green intellectual capital on competitive advantages of firms. Journal of Business Ethics 77, 271-286.
- Chen, Y., Chang, C., 2013. The determinants of green product development performance: green dynamic capabilities, green transformational leadership, and green creativity. Journal of Business Ethics 116, 107-119.
- Cheng, C., Shiu, E., 2012. Validation of a proposal instrument for measuring eco-innovation: An implementation perspective. Technovation 32, 329-344.
- Cheng, C. C., Yang, C. L., Sheu, C., 2014. The link between eco-innovation and business performance: a Taiwanese industry context. Journal of Cleaner Production, 64, 81-90.
- Chiarvesio, M., De Marchi, V., Di Maria, E., 2015. Environmental innovations and internationalization: Theory and Practices. Business Strategy and the Environment 24, 790-801.
- Christmann, P., 2000. Effects of 'best practices' of environmental management on cost advantage: the role of complementary assets. Academy of Management Journal 43, 663–680.
- Cohen, W. M., Levinthal, D. A., 1990. Absorptive capacity: a new perspective on innovation and

learning. Administrative Science Quarterly 35, 128-152.

- Cooper, R.G., Edgett, S.J., 2008. Maximizing productivity in production innovation. Research Technology Management 51, 47-58.
- Crabbé, A., Jacobs, R., Van Hoof, V., Bergmans, A., Van Acker, K., 2013. Transition towards sustainable material innovation: evidence and evaluation of Flemish case. Journal of Cleaner Production 56, 63-72.
- Cramer, J., Dral, P., Roes, B., 1991. Product information exchange about environmental aspects between producers. Ministry of Housing, Physical Planning and Environment, The Netherlands.
- Cramer, J., Schot, J., 1993. Environmental comakership among firms as a cornerstone in the striving for sustainable development, in: Fischer, K., Schot, J. (Eds), Environmental Strategies for Industry: International Perspectives on Research Needs and Policy Implications. Island Press, Washington, DC, pp. 311-328.
- Dalhammar, C., 2015. Industry attitudes towards ecodesign standards for improved resource efficiency. Journal of Cleaner Production 123, 155-166, http://dx.doi.org/10.1016/j.jclepro.2015.12.035.
- Dangelico, R. M., 2015. Green product innovations: where are and where we are going. Business Strategy and the Environment, 25, 560-576, http://dx.doi.org/IO.IOO2/bse.I886.
- De Jesús Pacheco, D. A., ten Caten, C. S., Jung, C. F., Ribeiro, J. L. D., Navas, H. V. G., Cruz-Machado, V. A., 2016. Eco-innovation determinants in manufacturing SMEs: Systematic review and research directions. Journal of Cleaner Production 142 (4), 2277-2287, https://doi.org/10.1016/j.jclepro.2016.11.049.
- De Marchi, V., 2012. Environmental innovation and R&D cooperation: empirical evidence from Spanish manufacturing firms. Research Policy 41, 614-623.
- De Medeiros, J. F., Ribeiro, J. L. D., Cortimiglia, M. N., 2014. Success factors for environmentally sustainable product innovation: a systematic literature review. Journal of Cleaner Production 65, 76-86.
- Del Brío, J. A., Junquera, B., 2003. A review of the literature on environmental innovation management in SMEs: implications for public policies. Technovation 23, 939-948.
- Del Río, P., Carrillo-Hermosilla, J., & Könnölä, T. (2010). Policy strategies to promote Eco-innovation. Journal of Industrial Ecology, 14(4), 541-557.
- Demirel, P., & Kesidou, E., 2011. Stimulating different types of eco-innovation in the UK: Government policies and firm motivations. Ecological Economics, 70(8), 1546-1557.
- Díaz-García, C., González-Moreno, A., Sáez-Martínez, F. J., 2015. Eco-innovation: insights from a literature review. Innovation Management, Policy and Practice, 17 (1), 6-23.
- Doran, J., Ryan, G., 2016. The importance of the diverse drivers and types of environmental innovation for firm performance. Business Strategy and the Environment 25, 102-119.

- Duhaime, R., Riopel, D., Langevin, A., 2001. Value analysis and optimization of reusable containers at Canada Post. Interfaces 31, 3-15.
- Eder, P., 2003. Expert inquiry on innovation options for cleaner production in the chemical industry. Journal of Cleaner Production 11 (4), 347-364.
- Ehrenfeld, J.R., 2008. Sustainability needs to be attained, not managed. Sustainability: Science, Practice and Policy 4, 1-11.
- EIO (Eco-Innovation Observatory), 2012. Methodological Report. Eco-Innovation Observatory. Funded by the European Commission, DG, Environment, Brussels.
- Ekins, P., 2010. Eco-innovation for environmental sustainability: concepts, progress and policies. International Economics and Economic Policy 7.2-3: 267-290.
- El Korchi, A., Millet, D., 2011. Designing a sustainable reverse logistics channel: the 18 generic structures framework. Journal of Cleaner Production 19, 588-597.
- Eltayeb, T., 2009. Adoption of green supply chain initiatives by ISO 14001 certified manufacturing firms in Malaysia: key drivers, outcomes, and moderating effect of relationship orientation. PhD Thesis, University Science Malaysia.
- European Commision, 2013. CIP eco-innovation first application and market replication projects. Frequently asked questions call 2013.
- Europa INNOVA, 2006. ThematicWorkshop, Lead Markets and Innovation, 29-30th June 2006, Munich, Germany.
- FAO (Food and Agricultural Organization of the United Nations), 2017. Food and sustainable agriculture. http://www.fao.org/sustainability/background/principle-5/en/
- Florida, R., 1996. Lean and Green: The Move to Environmentally Conscious Manufacturing. California Management Review 39 (1), 80-105.
- Frondel, M., Horbach, J., Reinnings, K., 2008. What triggers environmental management and innovation? Empirical evidence from Germany. Ecological Economics 66, 153-160.
- Frosch, R. A., 1994. Industrial ecology: minimizing the impact of industrial waste. Physics Today 47 (11), 63-68.
- Frosch, R., Gallopoulos, N., 1992. Towards and industrial ecology, in: Bradshaw, A.D., Southwood, R.,Warner, F. (Eds), The treatment and Handing of wastes. Chapman & Hall, London, pp. 269-292.
- Fussler, C., James, P., 1996. Eco-innovation: A Breakthrough Discipline for Innovation and Sustainability. Pitman Publishing, London.
- Garrod, B., Chadwick, P., 1996. Environmental management and business strategy: towards a new strategic paradigm. Futures 28 (1), 37-50.
- Ghisetti, C., Marzicchi, A., Montresor, S., 2015. The open eco-innovation mode. An empirical investigation of eleven European countries. Research Policy 44, 1080-1093.

- Ghisetti, C., Pontoni, F., 2015. Investigating policy and R&D effects on environmental innovation: a meta-analysis. Ecological Economics 118, 57-66.
- Ghisetti, C., Reinnings, K., 2014. Environmental innovations and profitability: how does it play to be green? An empirical analysis on the German innovation survey. Journal of Cleaner Production 75, 106-117.
- González-Torre, P. L., Adenso-Díaz, B., Artiba, H., 2004. Environmental and reverse logistics policies in European bottling and packaging firms. Journal of Production Economics 88, 95-104.
- Gottlieb, R., Smith, M., Roque, J., 1995. Greening or greenwashing?: the evolution of industry decision making, in: Gottlieb, R. (Ed.), Reducing Toxics. Island Press, Washington, DC, pp. 170-197.
- Griliches, Z., 1990. Patent statistics as economy indicators: a survey. Journal of Economic Literature 28 (4), 1661-1707.
- Groenewegen, P., Fischer, K., Jenkins, E. J., Schot, J., 1996. The greening of industry resource guide and bibliography. Island Press, Washington, DC.
- Guziana, B., 2011. Is the Swedish environmental technology sector 'green'? Journal of Cleaner Production 19, 827-835.
- Halbesleben, J. R. B., Novicevic, M. M., Harvey, M. G., Buckley, M. R., 2003. Awareness of temporal complexity in leadership of creativity and innovation: A competency-based model. Leadership Quarterly 14 (4/5), 433-454.
- Hamner, B., 2006. Effects of green purchasing strategies on supplier behavior, in: Sarkis, J. (Ed.), Greening the Supply Chain. Springer, London, pp.25-37.
- Hart, S.L., 1995. A natural-resource-based view of the firm. Academy of Management Review 20 (4), 986–1014.
- Hellström, T., 2007. Dimensions of environmentally sustainable innovation: the structure of ecoinnovation concepts. Sustainable Development 15 (3), 148-159.
- Henriksson, G., Akesson, L., Ewert, S., 2010. Uncertainty regarding waste handling in everyday life. Sustainability 2 (9), 2799-2813, http://dx.doi.org/10.3390/su2092799.
- Hojnik, J., Ruzzier, M., 2016a. The driving forces of process eco-innovation and its impact on performance: Insights from Slovenia. Journal of Cleaner Production 133, 812-825, https://doi.org/10.1016/j.jclepro.2016.06.002.
- Hojnik, J., Ruzzier, M., 2016b. What drives eco-innovation? A review of an emerging literature. Environmental Innovation and Societal Transitions 19, 31-41.
- Horbach, J., 2008. Determinants of environmental innovation- new evidence from German panel data sources. Research Policy 37 (1), 163-173.
- Iritani, D. R., Lopes Silva, D. A., Saavedra, Y. M. B., Grael, P. F. F., Ometto, A. R., 2014. Sustainable Strategies Analysis through Life Cycle Assessment: A Case Study in a Furniture Industry. Journal

of Cleaner Production 96, 308-318.

- Ishak, S. A., Hashim, H., Ting, T. S., 2016. Eco innovation strategies for promoting cleaner cement manufacturing. Journal of Cleaner Production 136, 133-149, http://dx.doi.org/10.1016/j.clepro.2016.06.022.
- Jänicke, M.,2012. Dynamic governance of clean-energy markets: how technical innovation could accelerate climate policies. Journal of Cleaner Production 22 (1), 50-59.
- Jelsma, J., 2016. Designing 'moralized' products, in: Verbeek, P. -P., Slob, A. (Eds.), User Behaviour and Technology Development. Springer, Netherlands, pp.221-231.
- Johnstone, N., Hascic, I., Popp, D., 2010. Renewable Energy Policies and Technological Innovation: Evidence Based on Patent Counts. Environmental Resource Economics 45, 133-155.
- Jolly, A., Philpott, J., 2004. A Handbook of Intellectual Property Management: Protecting, Developing and Exploiting Your IP Assets. Kogan Page Publishers (Ed.), London.
- Juul, S., 2012. Food Waste: a Global Tragedy. http://tastethewaste.com/article/20120123-Food-waste-a-global-tragedy- (accessed 9 October 2017).
- Kemp, R., Arundel, A., 1998. Survey indicators for Environmental Innovation. IDEA report. Step group. Oslo.
- Kemp, R., 2009. Measuring eco-innovation: results from the MEI project. Global Forum on Environment on Eco-innovation, 4-5 Nov. Organisation for Economic Co-operation and Development (OECD), Paris.
- Kemp, R., Pearson, P., 2007. Final Report MEI Project about Measuring Eco-innovation. Project deliverable 15. http://www.oecd.org/env/consumption-innovation/43960830.pdf.
- Kemp, R., Pearson, P., 2008. Policy brief about measuring eco-innovation and Magazine/Newsletter articles. Project deliverable 17. https://cordis.europa.eu/docs/publications/1245/124548931-6_en.pdf.
- Kesidou, E., Demirel, P., 2012. On the drivers of eco-innovations: empirical evidence from the UK. Research Policy 41, 862-870.
- Lacerda, J. S., Van den Bergh, J. C. J. M., 2014. International Diffusion of Renewable Energy Innovations: Lessons from the Lead Markets for Wind Power in China, Germany and USA. Energies 7, 8236-8263.
- Langley, J., Turner, N., Yoxall, A., 2011. Attributes of packaging and influences on waste. Packaging, Technology and Science 24, 161-175.
- Lanjouw, J. O., Mody, A., 1996. Innovation and the International Diffusion of Environmentally Responsive Technology. Research Policy 25, 549-571.
- Läpple, D., Renwick, A., Thorne, F., 2015. Measuring and understanding the drivers of agricultural innovation: Evidence from Ireland. Food Policy 51, 1-8.

- Li, X., Hamblin, D., 2016. Factors impacting on cleaner production: case of studies of Chinese pharmaceutical manufacturers in Tianjin, China. Journal of Cleaner Production 131, 121-132, https://doi.org/10.1016/j.jclepro.2016.05.066.
- Lindh, H., Williams, H., Olsson, A., Wilkström, F., 2016. Elucidating the indirect contributions of packaging to sustainable development - a terminology of packaging functions and features. Packaging, Technology and Science 29 (4/5), 225-246.
- Löfgren, M., 2005. Winning at the first and second moment of truth: an exploratory study. Management Service Quality 15 (1), 102-115.
- Loorbach, D., Wijsman, K., 2013. Business transition management: exploring a new role for business in sustainability transitions. Journal of Cleaner Production 45, 20-28.
- Macron, A., de Medeiros, J. F., Ribeiro, J. L. D., 2017. Innovation and environmentally sustainable economy: Identifying the best practices developed by multinationals in Brazil. Journal of Cleaner Production 160, 83-97, https://doi.org/10.1016/j.jclepro.2017.02.101.
- Martin, M., Williams, J. D., Clark, M., 2006. Social, cultural and structural influences on household waste recycling: a case study. Resources, Conservation and Recycling 48, 357-395.
- Matos, S., Silvestre, B. S., 2013. Managing stakeholder relations when developing sustainable business models: the case of the Brazilian energy sector. Journal of Cleaner Production 45, 61-73.
- Montalvo, C., 2003. Sustainable production and consumption systems-cooperation for change: assessing and simulating the willingness of the firm to adopt/develop cleaner technologies. The case of the In-Bond industry in northern Mexico. Journal of Cleaner Production 11, 411-426.
- Montalvo, C., 2008. General Wisdom concerning the factors affecting the adoption of cleaner technologies: a survey 1990-2007. Journal of Cleaner Production 16S1, S7-S13.
- Moore, R., 2005. Reverse logistics-the least used differentiator. UPS Supply Chain Solutions, Georgia, USA.
- Morioka, S. N., de Carvalho, M. M., 2016. A systematic literature review towards a conceptual framework for integrating sustainability performance into business, Journal of Cleaner Production, 136 (Part A), 134-146, https://doi.org/10.1016/j.jclepro.2016.01.104
- Naffziger, D. W., Ahmed, N. U., Montagno, R., 2003. Perceptions of environmental consciousness in US small businesses: An empirical study. SAM Advanced Management Journal 68, 23-32.
- Nesta, L., Vona, F., Nicolli, F., 2014. Environmental policies, competition and innovation in renewable energy. Journal of Environmental Economics and Management 67, 396-411.
- Nicolli, F., Vona, F., 2016. Heterogeneous policies, heterogeneous technologies: The case of renewable energy. Energy Economics 56, 190-204.
- Niinimäki, K., Hassi, L., 2011. Emerging design strategies in sustainable production and consumption of textiles and clothing, Journal of Cleaner Production 19 (16), 1876-1883.

- Norberg-Bohm, V., 1999. Stimulating 'green' technological innovation: an analysis of alternative policy mechanisms. Policy Sciences 32 (1), 13-38.
- O'Connor, G. C., Ayers, A. D., 2005. Building a radical innovation competency. Research Technology Management 48 (1), 23-31.
- OECD (Organisation for Economic Co-operation and Development), 2005. Oslo manual guidelines for collecting and interpreting innovation data. Organisation for Economic Co-operation and Development: Statistical Office of the European Communities, Paris.
- OECD (Organisation for Economic Co-operation and Development), 2012. Perspectivas ambientales de la OCDE hacia 2050. Consecuencias de la inacción. https://www.oecd.org/env/indicators-modelling-outlooks/49884278.pdf.
- Oltra, V., Kemp, R., de Vries, F., 2008. Patents as a measure for eco-innovation. Report for MEI project, deliverable 9.

https://pdfs.semanticscholar.org/6bd8/a3add30a083d808c6ff54de877eba1b19a43.pdf

- Oltra, V., Saint Jean, M., 2009. Sectorial systems of environmental innovation: an application to the French automotive industry. Technological Forecasting and Social Change 76 (4). 567-583.
- Peng, X., Liu, Y., 2016. Behind eco-innovation: Managerial environmental awareness and external resource acquisition. Journal of Cleaner Production 139, 347-360, https://doi.org/10.1016/j.jclepro.2016.08.051.
- Petticrew, M., Roberts, H., 2012. Systematic Reviews in the Social Sciences. A Practical Guide. Blackwell (Ed.), Oxford, UK.
- Porter, M., Van der Linder, C., 1995. Green and competitive: ending the stalemate. Harvard Business Review 73 (5), 119-134.
- Piedra-Muñoz, L., Godoy-Durán, A., Giagnocavo, C., 2017. How to Improve Water Usage Efficiency? Characterization of Family Farms in a Semi-Arid Area. Water, 9, 785, https://doi.org/10.3390/w9100785.
- Piedra-Muñoz, L., Vega-López, L.L., Galdeano-Gómez, E., Zepeda-Zepeda, J.A., 2018. Drivers for efficient water use in agriculture: An empirical analysis of family farms in Almería (Spain). Experimental Agriculture, 54 (1), 31-44.
- Pigosso, D. C., Zanette, E. T., Guelere Filho, A., Ometto, A. R., Rozenfeld, H., 2010. Ecodesign methods focused on remanufacturing. Journal of Cleaner Production 18 (1), 21-31.
- Plumb, A., Downing, P., Parry, A., 2013. Final Report: A Consumer Attitudes to Food Waste and Food Packaging. WRAP, UK. http://www.wrap.org.uk. (accessed 3 September 2015).
- Rajala, R., Westerlund, M., Lampikoski, T., 2016. Environmental sustainability in industrial manufacturing: reexamining the greening of Interface's business model. Journal of Cleaner Production 115, 52-61, http://dx.doi.org/10.1016/j.jclepro.2015.12.057.

- Rennings K., 2000. Redefining innovation eco-innovation research and the contribution from ecological economics. Ecological Economics 32: 319-332
- Rennings, K., Zwick, T. (Eds.), 2003. Employment Impacts of Cleaner Production, ZEW Economic Studies, Bd.21, Heidelberg.
- Remacha, M., 2017. Empresa y objetivos de desarrollo sostenible. Cuadernos de la Cátedra CaixaBank de Responsabilidad Social Corporativa 34, 1-27.
- Rodríguez, J., Wiengarten, F., 2017. The role of process innovativeness in the development of environmental innovativeness capability. Journal of Cleaner Production 142, 2423-2434.
- Rogers, D. S., Tibben-Lembke, R. S., 1998. Going Backwards Reverse Logistics Trend and Practices. University of Nevada, Center for Logistics Management, Reno.
- Roscoe, S., Cousins, P. D., Lamming, R. C., 2015. Developing eco-innovations: a three-stage typology of supply networks. Journal of Cleaner Production 112, 1948-1959, http://dx.doi.org/10.1016/j.jclepro.2015.06.125.
- Rosenau, W. V., Twede, D., Mazzeo, M. A., Singh, S. P., 1996. Returnable/reusable logistical packaging a capital budgeting investment decision framework. Journal of Business Logistics 17, 139-165.
- Scarpellini, S. Aranda, A., Aranda, J., Llera, E., Marco, M., 2012. R&D and eco-innovation: opportunities for closer collaboration between universities and companies through technology centers. Clean Technologies and Environmental Policy 14 (6), 1047-1058.
- Segarra-Oña, M. V., Peiró-Signes, A., 2014. Environmental eco-orientation determinants at the Spanish tile industry. Dyna 89 (2), 220-228.
- Shrivastava, P., 1995. Environmental technologies and competitive advantage. Strategic Management Journal 16, 183–200.
- Shrivastava, P., 1996. Greening business: Profiting the corporation and the environment.. Thomson Executive Press (Ed.), Cincinnati, Ohio.
- Sierra-Pérez, J., López-Forniés, I., Boschmonart-Rives, J., Gabarell, X., 2016. Introducing eco-ideation and creativity techniques to increase and diversify the applications of eco-materials: The case of cork in the building sector. Journal of Cleaner Production 137, 606-616.
- Silva, D. A. L., Renó, G. W. S., Sevegnani, G., Sevegnani, T. B., Truzzi, O. M. S., 2013. Comparison of disposable and returnable packaging: a case study of reverse logistics in Brazil. Journal of Cleaner Production 47, 377-387.
- Stock, J. R., 1992. Reverse Logistics. Council of Logistics Management. Oak Brook, Illinois.
- Stock, J. R., Lambert, D. M., 2001. Strategic Logistics Management. McGraw-Hill, New York.
- Tamayo-Obergozo, U., Vicente-Molina, M. A., Villarreal-Larrinaga, O., 2017. Eco-innovation strategic model. A multiple-case study from a highly eco-innovative European region. Journal of Cleaner Production 142, 1347-1367.

- Theyel, G., 2000. Management practices for environmental innovation and performance. International Journal of Operations and Production Management 20 (2), 249-266.
- Tranfield, D., Denyer, D., Smart, P. (2003). Towards a methodology for developing evidence-informed management knowledge by means of systematic review. British Journal of Management, 14, 207– 222.
- Triguero, A., Moreno-Mondéjar, L., Davia, M. A. (2013). Drivers of different types of eco-innovation in European SMEs. Ecological economics, 92, 25-33.

Tseng, M. L., Wang. R., Chiu, A. S. F., Geng, Y., Lin, Y. H., 2013. Improving performance of green innovation practices under uncertainty. Journal of Cleaner Production 40, 71-82, https://doi.org/10.1016/j.jclepro.2011.10.009.

- Tibbs, H., 1992. Industrial ecology- an agenda for environmental management. Pollution Prevention Review Spring, 167-180.
- Twede, D., Clarke, R., 2005. Supply chain issues in reusable packaging. Journal of Marketing Channels 12, 7-26.
- Van Hemel, C., Cramer, J., 2002. Barriers and stimuli for ecodesign in SMEs. Journal of Cleaner Production 10, 439-453.
- Wagner M, Llerena P, 2011. Eco-innovation through integration, regulation and cooperation: comparative insights from case studies in three manufacturing sectors. Industry and Innovation, 18(8), 747-764.
- Wever, R., Vogtländer, J., 2014. Design for the value of sustainability, in: Handbook of Ethics, Values, and Trchnological Design. Springer, Netherlands, pp.1-31.
- Williams, H., Medhurst, J., Drew, K., 1993. Corporate strategies for a sustainable future, in: Fischer, K., Schot, J. (Eds), Environmental Strategies for Industry: International Perspectives on Research Needs and Policy Implications. Island Press, Washington, DC, pp. 117-143.
- Wilkström, F., Williams, H., Venkatesh, G., 2016. The influence of packaging attributes on recycling and food waste behaviour- An environmental comparison of two packaging alternatives. Journal of Cleaner Production 137, 895-902.
- Ye, X., Zhang, X., 2013. PLM for multiple lifecycle products: concepts, terminologies, processes for collaborative information management (Master thesis). KTH Royal Institute of Technology, Stockholm, Sweden.
- Zailani, S., Jeyaraman, K., Vengadasan, R., Premkumar, R., 2012. Sustainable supply chain management (SSCM) in Maylasia: A survey. International Journal of Production Economics 140, 330-340.